Author's response to reviews

Title: Prognostic Impact of Array-based Genomic Profiles in Esophageal Squamous Cell Cancer

Authors:

Ana Carneiro (Ana.Carneiro@med.lu.se)
Anna Isinger (Anna.Isinger@med.lu.se)
Anna Karlsson (Anna_F.Karlsson@med.lu.se)
Jan Johansson (Jan.Johansson@med.lu.se)
Goran Jonsson (Goran_B.Jonsson@med.lu.se)
Par-Ola Bendahl (Par-Ola.Bendahl@med.lu.se)
Dan Falkenbach (Dan.Falkenbach@HelsingborgsLasarett.se)
Britta Halvarsson (Britta.Halvarsson@med.lu.se)
Mef Nilbert (Mef.Nilbert@med.lu.se)

Version: 3 Date: 10 March 2008

Author's response to reviews:

Re: MS: 1998581436164416
¿Prognostic Impact of Array-based Genomic Profiles in Esophageal Squamous Cell Cancer¿.

Dear Editor,

Thank you for accepting our manuscript for publication in BMC Cancer.

We have clarified, in the Materials and Methods section, that written informed consent was obtained from the patients.

On behalf of the authors, sincerely yours

Mef Nilbert and Ana Carneiro